Author’s response to reviews

Title: Health care seeking delay among pulmonary tuberculosis patients in North West zone of Tigrai region, North Ethiopia

Authors:
Haileselasie Alema (haileselasieb@yahoo.com)
Sisay Hailemariam (sisasgedom@gmail.com)
Kebede Misgina (kebede.haile82@gmail.com)
Meresa Woldu (mere2010ec@yahoo.com)
Yosef Gebregergis (yosefsibhatu@gmail.com)
Genet Mekonen (genetkidane971@yahoo.com)
Kahsay Gebremedhin (kahsay93@gmail.com)

Version: 3 Date: 05 Mar 2019

Author’s response to reviews:

Comment 1

please make professional language editing service

Response to comment 1

We made revision on the manuscript wording, grammar and punctuation.

Comment 2
The Availability of data and materials section refers to the raw data used in your study and presenting tables and figures is not sufficient to state that all data is contained within the manuscript and additional files.

Response to comment 2

The datasets used and/or analysed during the current study available from the corresponding author on reasonable request.

Comment 3

In the section 'Funding', please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response to comment 3

Despite reviewing this research paper prior to submission for publication, the partial funder has no any role in study design, data collection, data analysis and interpretation of the data and writing the manuscript.

Comment 4

We note that you have included the initials 'KB' in the Authors' Contribution section of your manuscript. However, these initials do not correlate to any of the authors listed. Please clarify whether 'KB' is referring to an author that has not been listed on the author list.

Response to comment 4
Comment accepted and corrected. sorry, it was typing error

Comment 5

Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case.

Response to comment 5

All the authors have read and approved the manuscript for publication ready.